Author's response to reviews

Title: Measuring inequalities in the distribution of health workers: the case of Tanzania

Authors:

Michael A Munga (michaelmunga@yahoo.com)
Ottar Maestad (ottar.mestad@cmi.no)

Version: 3 Date: 12 November 2008

Author's response to reviews: see over
To the Editor In chief.
Human Resources for Health (BioMedCentral) 12/11/08

Re: Submission of the manuscript titled: “Measuring inequalities in the distribution of health workers: the case of Tanzania” (third draft)

Authors: Michael A Munga and Ottar Maestad

Kindly refer to the above caption.

On behalf of my co-author, I hereby submit a revised manuscript with the above title to the journal of human resources for health.

We have addressed the editorial comments as listed below:

1. We have provided full and correct authors’ names and their institutional affiliation as suggested
2. We have in this version, removed section numbers for the whole manuscript
3. We have removed all footnotes and incorporated the notes in the main text. The references used in the footnotes have been transferred to the reference list at the end of the paper.
4. We have checked, corrected and ensured that the references adhere to the specified referencing guidelines
5. We have minimised borders around figures. In this version, figure title and numbers are not embedded in the figures
6. We have uploaded each figure as a separate file as suggested and;
7. We have moved all tables to the end of the text.

Thank you for considering our manuscript for publication in your esteemed journal

Regards,
Michael A. Munga (Corresponding author)